



# **A Review of EV Battery Utilization in Demand Response Considering Battery Degradation in Non-Residential Vehicle-to-Grid Scenarios**

Andre Leippi <sup>1,2</sup>, Markus Fleschutz <sup>1,2</sup> and Michael D. Murphy <sup>1,\*</sup>

- <sup>1</sup> Department of Process, Energy and Transport Engineering, Munster Technological University, T12 P928 Cork, Ireland; andre.leippi@mycit.ie (A.L.); markus.fleschutz@mycit.ie (M.F.)
- <sup>2</sup> Institute of Refrigeration, Air-Conditioning, and Environmental Engineering, Karlsruhe University of Applied Sciences, 76133 Karlsruhe, Germany
- \* Correspondence: michaeld.murphy@mtu.ie

Abstract: Integrating fleets of electric vehicles (EVs) into industrial applications with smart grids is an emerging field of important research. It is necessary to get a comprehensive overview of current approaches and proposed solutions regarding EVs with vehicle-to-grid and smart charging. In this paper, various approaches to battery modeling and demand response (DR) of EV charging in different decentralized optimization scenarios are reviewed. Modeling parameters of EVs and battery degradation models are summarized and discussed. Finally, optimization approaches to simulate and optimize demand response, taking into account battery degradation, are investigated to examine the feasibility of adapting the charging process, which may bring economic and environmental benefits and help to alleviate the increasing demand for flexibility. There is a lack of studies that comprehensively consider battery degradation for EV fleets in DR charging scenarios where corresponding financial compensation for the EV owners is considered. Therefore, models are required for estimating the level of battery degradation endured when EVs are utilized for DR. The level of degradation should be offset by providing the EV owner with subsidized or free electricity provided by the company which is partaking in the DR. This trade-off should be optimized in such a manner that the company makes cost savings while the EV owners are compensated to a level that is at least commensurate with the level of battery degradation. Additionally, there is a lack of studies that have examined DR in smart grids considering larger EV fleets and battery degradation in multi-criteria approaches to provide economic and environmental benefits.

**Keywords:** multi-objective optimization; electric vehicle fleet; electric vehicle charging; industrial demand response; vehicle-to-grid; smart grid; battery degradation

# 1. Introduction

In an effort to achieve climate neutrality by 2050, the European Commission has decided to design the European climate, energy, land use, transport, and taxation policies in such a way that net greenhouse gas emissions can be reduced by at least 55% by 2030 compared to 1990 levels. For this purpose, the Climate Change Regulation formulates the objectives in the transport sector that complement emissions trading. A key milestone in the development of the transport sector will be stricter  $CO_2$  emission standards for passenger cars and light commercial vehicles, which will accelerate the transition to zero-emission mobility, as the average annual emissions of new vehicles will have to be 55% lower from 2030 and 100% lower from 2035 compared to 2021 levels [1]. Therefore, all new cars registered from 2035 onwards will be zero-emission. It will also ensure that drivers can charge or refuel their vehicles on a reliable network across Europe by increasing charging capacities in line with the sales of zero-emission vehicles and by setting up charging and refueling stations at regular intervals on major highways [2]. The substitution of



Citation: Leippi, A.; Fleschutz, M.; Murphy, M.D. A Review of EV Battery Utilization in Demand Response Considering Battery Degradation in Non-Residential Vehicle-to-Grid Scenarios. *Energies* 2022, 15, 3227. https://doi.org/ 10.3390/en15093227

Academic Editor: Enrique Romero-Cadaval

Received: 16 March 2022 Accepted: 25 April 2022 Published: 28 April 2022

**Publisher's Note:** MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



**Copyright:** © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). internal combustion engine vehicles (ICEVs) with EVs provides a promising opportunity for countries' transportation systems to meet  $CO_2$  reduction commitments and thus mitigate climate change and air pollution. Therefore, it is important to study the global emissions of EVs compared to ICEVs from a whole life cycle perspective, e.g., using the widely applied life cycle analysis method [3,4].

In [5], current and future electric vehicles were studied in terms of environmental impact. The results show that in the future scenario, EVs are generally preferable to PHEVs and ICEVs in the context of environmental impact for the same lifetime and vehicle characteristics. In general, previous studies show that EVs are suitable, at the local level, to create less pollutant emissions than ICEVs in the transport sector [4-7]. As a result, governments in many European countries are promoting electric mobility, which has already led to a substantial increase in electric vehicles over the last several years. The European market for EVs experienced significant growth in 2020. More than 1.36 million new electric passenger cars, including battery electric vehicles (BEVs) and plug-in hybrids (PHEVs), were sold across the region. This represents an increase of 143% compared to 2019. In this paper, for the sake of simplicity, battery electric vehicles (BEVs) are referred to as electric vehicles (EVs). While the share of EVs in the total passenger new car market was around 2% in 2018, it increased from 4% to more than 11% between 2019 and 2020 [8]. In the European core regions, the EV market grew by 147% year-on-year in the first half of 2021 to a total of 336.000 BEVs sold. PHEV sales increased by 248% to 350.000 units. Germany dominates the market in Europe with 44% of all BEV sales and 47% of PHEV sales in the five European core markets in the first half of the year [9].

Due to the ongoing trend, it is essential to deal with the increasing number of EVs, as both the electrical load and the demand on the electricity markets increase. According to the Global EV Outlook 2021 [10], the global EV fleet across all modes (excluding two- and three-wheelers) will increase from over 11 million in 2020 to nearly 145 million vehicles by 2030. This corresponds to an annual average growth rate of almost 30%. Furthermore, EVs will account for approximately 7% of the road vehicle fleet in 2030, and EV sales will reach almost 15 million in 2025 and over 25 million in 2030, representing 10% and 15% of all road vehicles sold, respectively. As electricity generation based on renewable resources requires flexibility on the consumer side, the electricity market is expected to adopt more dynamic electricity prices to incentivize demand response. To take advantage of this price volatility, companies could implement vehicle-to-grid (V2G) and flexible smart charging in the manner of a service provider. A major challenge is to integrate fleets of EVs into smart grids, as stakeholders need to be able to manage the charging and discharging of EV fleets in a way that achieves economic, technological, and environmental benefits. With the use of renewable energy generation and different price signals in electricity markets mainly due to the synergistic use of renewables and EVs, certain stakeholders could achieve several benefits. In terms of an industrial or non-residential setting, these stakeholders are the industrial-based electricity-consuming companies that may be able to utilize EVs as mobile and low-cost electricity storage. The EV owner (employee of the company) can be compensated for providing their vehicle batteries as temporary storage. The electricity grid operator will benefit, as the penetration of intermittent renewable energy sources will be more easily facilitated. EV owners' concerns about battery degradation should be appropriately addressed so that an optimal trade-off between increasing economic performance for demand response for the company and financially compensating EV owners at a level that is at least equal to the battery degradation endured. To represent the trade-off between these stakeholders in the industrial setting, who would not necessarily collaborate, multi-criteria optimization problems can be constructed to simulate one or more (partly competing) objectives. Namely, minimizing the company's electricity costs, CO<sub>2</sub> emissions, and EV battery degradation costs.

Therefore, effective methods are needed to improve the economic and environmental use of the growing supply of EVs. As such, this study focuses on critically evaluating the published literature dealing with the following three main components: demand response in non-residential settings, optimization of different and competing objectives, and the analysis of battery degradation in EV charging scenarios. The inclusion of battery degradation is essential in achieving a more realistic modeling environment. Many EV owners do not want their EVs used for demand response participation as the many charging and discharging cycles that take place over a given period will ultimately reduce the capacity of their EV battery due to degradation, even if a short-term economic gain is achieved. In this context, Huang et al. [11] investigated which preferences Dutch EV drivers have for participation in V2G contracts. The results show that Dutch EV drivers are most concerned with discharge cycles. It was also found that Dutch EV drivers prefer a higher monthly remuneration for providing their EV and a higher guaranteed minimum battery state of charge (SOC). The SOC is a percentage value and is calculated as the ratio of the net capacity to be discharged and the nominal capacity of the battery [12].

Geske and Schumann [13] also investigated the willingness to participate in V2G using a representative sample of vehicle users in Germany. Similar to [11], they found that range anxiety and minimum range emerged as the most important factors affecting vehicle users' willingness to participate in V2G. Furthermore, their analysis showed that high remuneration is not necessary for significant participation rates if the V2G concept is convincing and provides enough room for unpredictable and predictable mobility demand. Most EV owners and grid operators cannot quantify the trade-off that exists between utilizing the untapped capacity of EVs for smart grid demand response utilization and the resulting reduced capacity of the EV batteries as a result of increased utilization and therefore battery degradation. The different objectives should be simulated depending on the individual interests of the stakeholders in order to control the charging and discharging of EVs in the best way to optimize economic and environmental benefits.

The contribution of this work is to provide an overview of the current state of battery modeling and optimization strategies in V2G and smart charging scenarios of EVs. In particular, the interaction of the three main components mentioned above is investigated to identify gaps in the utilization of EV batteries in an industrial DR for future research. This review is structured as follows. Section 2 briefly explains the methodology used in the literature review. Section 3 explores research studies that focus on non-residential demand response. Section 4 summarizes different approaches to battery modeling and charging parameters in different EV charging scenarios. Section 5 focuses on battery degradation and different battery degradation models. Section 6 presents modeling approaches that already consider battery degradation as well as different optimization targets. After a presentation of the current state of the art in the field of optimal control of EV charging and discharging in the context of demand response and battery degradation, Section 7 discusses the identified gaps in the field of EV charging. Section 8 provides a concise conclusion to this review.

## 2. Methodology

We first reviewed the sections on non-residential DR, EV battery modeling, and battery degradation. Then, the interactions between these components were examined to identify gaps in the use of EV batteries for economic and environmental benefits. In conducting this review, we proceeded in three steps. In the first step, we formulated preliminary research topics or research questions: the goal was to broadly examine the status of EV battery use in demand response, taking into account battery degradation.

The second step was to select the digital libraries of the studies and the timeframe. The following digital libraries were searched for the review: Google Scholar, MDPI, Scopus, IEEE Xplorer, ScienceDirect, Springer, and Elsevier. For the databases, we used a set of including and excluding keywords. The following two exclusionary criteria were generally used: we limited the search to English-language references and to those published since 2013. Due to the relatively innovative and novel concept of V2G regarding the integration of electric vehicles into smart grids, we chose this time period because most electric vehicles on the market today, as well as many EVs coming to market in the next few years, are not bidirectional or V2G capable [14]. Nowadays, there are still only a few EVs V2G capable such as the Nissan Leaf [15], the Mitsubishi Outlander [16], or the Kia Soul [17]. Nevertheless, the development is heading in the direction of a more grid-interactive network system. For example, all Volkswagen ID models with the 77 kWh battery will be enabled for V2G technology in the future [18].

The third step was to select and analyze the literature. We searched the above databases for studies, reviews, summaries, keywords, and full texts for the terms such as "Electric Vehicle \*", "Vehicle-to-Grid", "Demand Response", "Smart Charging", "Smart Grids" and their abbreviations. For Table 1 ("Battery modeling and charging parameters found in the literature"), the selection criteria were broad and included studies published on EV charging or charging strategies to provide an overview of the state of the art on battery modeling and charging parameters. Optionally, keywords such as "V2G", "Renewable Energy", "Battery Degradation" or "CO<sub>2</sub> Emissions" were included in the search.

For Table 2 ("Battery degradation models"), empirical, semi-empirical, and electrochemical/mechanical degradation models were reviewed to investigate the impact of different influencing parameters from different perspectives. The focus was on studies addressing "cyclic battery degradation"; studies addressing calendar degradation were not explicitly searched.

For Table 3 ("Optimization scenarios in smart grids considering battery degradation"), keywords such as "Multi-Objective Optimization", "Optimization", "Batteries" and "Energy arbitrage" were included in addition to the keywords from the search in Table 2. In addition, only studies that included any form of cyclic battery degradation for energy storage (EV or BESS) were included. Therefore, the keywords "degradation" or "battery degradation" were required. BESSs were also included as it was found that there were few studies in this area. The keyword "Demand Response" was only optionally included in the search for the same reason. Although some studies consider price-based DR with different tariffs and different prices or incentive-based DR, a large gap was apparent in this field when the term "Demand Response" was added (especially in combination with "Industry" or "Commercial").

# 3. Non-Residential Demand Response

In this work, we focus on DR for non-residential buildings where employees work. This is typically the case in the industrial and commercial sectors including industrial and large commercial firms as well as public sector organizations (see also [19]).

DR is the provision of demand-side flexibility by a final electricity consumer to the electricity system either by reacting voluntarily to price signals (price-based DR) or reacting to specific requests (incentive-based DR) [20]. Driven by digitalization and the trend towards decentralization within the energy transition, DR is a cost-effective alternative to other flexibility sources, such as storage or grid expansion [19,21]. Consequently, as the electricity system's demand for flexibility is expected to quadruple by 2050, DR is becoming increasingly important [22]. DR is a key element for the integration of fluctuating renewable energy sources in the electricity system, which is a major challenge in the renewable energy transition [21,23]. The high relevance of DR is reflected by intense research activities [24] and numerous research projects funded at European [25–27] and national levels [28–30]. The research on DR can be classified by the sectors of electricity users: the residential, commercial, and industrial sectors [19]. However, when only cross-sectional technologies are considered, industrial companies can be grouped with large commercial companies (e.g., [31]). Similarly, small commercial businesses can be grouped with the residential sector due to their similar electricity consumption (e.g., [32]). This is also reflected in the electricity end-user categorization [19]. Although the theoretical potential of the three sectors is estimated to be similar in magnitude [33], the commercial sector gains less scientific attention than the others [19,34].

Traditionally, only the industrial sector has been able to participate in DR schemes [35]. Due to large equipment dimensions and the existence of advanced IT infrastructure, indus-

trial DR can often be applied without high investments and has therefore been especially promising [36]. However, since the advent of flexibility aggregators, the commercial and residential sectors also have access to DR programs, albeit only indirectly. Other barriers to DR deployment are expected to be gradually removed as the demand for flexibility in the power system increases. Moreover, by installing on-site renewable energy sources, more and more electricity consumers become prosumers. Assuming a price gap between purchasing and feed-in electricity price, these prosumers can directly benefit from their demand-side flexibility that enables them to optimize their energy demand and feed-in profile and reduce electricity costs as a result.

Over the last three decades, research in the field of non-residential DR potentials and modeling has been steadily intensified, resulting in a large and growing body of literature.

Most of the studies focus on industrial production processes (e.g., [37–40]) while fewer on cross-sectional technologies such as distributed energy resources [41] (e.g., combined heat and power [42–44], stationary battery storage [45]), air separation [46] and heat upgrading technologies (e.g., electric heat pumps [31,47]) in industrial and commercial buildings. For some recent reviews on industrial DR, the reader is referred to [19,24,34,36,48,49].

Despite a large number of studies on DR that incorporate EVs, the literature lacks a focus on the potential of utilizing the employee's EV batteries to shape the demand profile of an employer.

A recent study by Shahnewaz Siddiquee et al. [48] conducted a comprehensive literature review on research trends, current implementation state, and adoption barriers of industrial DR. The study particularly highlighted significant potentials in large industries such as aluminum, cement, and food. As the main barriers to the application of industrial DR, the study identified the lack of access to information, technical constraints, but also lacking financial mechanisms, and DR promoting policies. Here, the lacking information access was primarily an insufficient knowledge about the available flexibility. To solve this problem of quantifying the specific DR potential, model-based approaches can be used which are usually based on simulation, metaheuristics, or mathematical optimization [50].

# 4. Battery Modeling

In the literature, there are different approaches to model individual EV or EV fleets to take advantage of the multiple benefits of V2G and smart charging. In energy management systems, the charging and discharging process of an EV is characterized by the dynamics of its battery state. Batteries are modeled by a dynamic SOC equation. Thereby, the EV battery is mostly simulated as a temporarily available energy storage system that is subject to technical constraints such as capacity and temporal restrictions. In general, the modeling of EV batteries and their charging characteristics can be divided into two basic categories. One is a model where the battery of each vehicle is modeled individually. Here, EV aggregators are often used to control and manage the SOC, charge or discharge, and communication of all EVs to represent them on an aggregate level for the power exchange between the vehicle, the grid, and the demand. While optimizing the charging/discharging of individual EVs is more complex and time-consuming, it can still address the needs and potential V2G contracts of EV owners. Flexible and practical strategies can be integrated such as desired SOCs or minimum EV range at specific times for individual EV owners. The other is a model in which the batteries of an often large number of individual EVs in a vehicle fleet are combined into a single aggregate battery, which clusters the EVs and represents them as one virtual battery. In this case, this model usually has only a single state of charge variable and enables efficient optimizations with reduced computational expenses.

Table 1 summarizes different charging and objective implementations, the type of battery modeling, the number of EVs considered, and various battery modeling parameters such as charging and discharging efficiencies, SOC restrictions, and C-rates of different EV charging scenarios. In addition, we investigated which studies have taken into account the following aspects: the possibility of V2G, the integration of renewable energy, and the consideration of  $CO_2$  emissions and battery degradation. The C-rate is a measure of

the maximum rate at which a battery is charged or discharged in relation to its maximum capacity. A C-rate of 1 means that the battery can be completely discharged in 60 min, and 0.5 C equals to 120 min. In the 23 studies reviewed, as well as in most charging control strategies in the literature, electric vehicles are modeled individually. Only a few studies use a single aggregated battery with common variables or parameters, as these represent limited conditions for charge control. Thus, the state and status of each vehicle's battery can be integrated as a direct constraint for charging optimization. The studies reviewed focused primarily on the analysis of:

- Charging optimization strategies: There are various strategies that regulate the charging behavior of EVs. These include uncontrolled and controlled charging. In the case of uncontrolled charging, the simulations (Table 1) often assume a charging behavior of "as fast as possible". In this behavior, the EV starts charging immediately when it is connected to the grid and stops charging as soon as a certain SOC is reached (e.g., a fully charged battery) or when the vehicle leaves the charging station. In coordinated charging, the EVs' charging loads are intelligently controlled and managed (smart charging) to minimize charging costs through a dynamic pricing policy.
- Benefits of additional services that EVs can provide: These are consistent with the benefits of the V2G concept, including reducing grid congestion concerns (peak shaving and load balancing through valley filling). They offer ancillary services such as voltage and frequency regulation through cheap and fast energy storage, possible support for renewable energy sources, and use as spinning reserves [51,52].
- Increasing the techno-economic potential of EV microgrids or smart grids including energy arbitrage: Aside from reducing utility operating costs, revenues can be generated using EVs. This includes V2G energy arbitrage, where electricity is bought from the grid and temporarily stored in the EV when the electricity price is low. Then the excess energy is sold to the grid when the price of electricity is considered high. Another way to make profits is the so-called "Beneficial V2G". This involves deliberately discharging and then recharging a battery during the performance of V2G services to reduce overall battery degradation [53,54]. Therefore, the reciprocal effect of several factors such as minimizing energy costs, accounting for battery degradation, CO<sub>2</sub> emissions, and, in this context, the inclusion of renewable energy sources should be considered to weight these competing objectives.

Study	V2G	RES	Degr.	CO <sub>2</sub>	Battery Modeling	Number (Dis)Charge of EVs Efficiency		SOC Limits	(Max) C-Rate	
Malya et al. (2021) [54]	$\checkmark$	×		×	1EV	1	0.85-(up to 1)	$\geq 10\%$	22 kW (1 C)	
Cardoso et al. (2014) [55]	v		v		AB	1080	0.9	$\geq 20\%$	11 kW (0.45 C)	
Di Giogio et al. (2014) [56]	v	×	×	×	XB	63	0.98	20-90%	19.8 kW (0.9 C)	
Liu et al. (2015) [57]	×	$\checkmark$	×	×	ХВ	60	n/a	Objective SOC 85%	0.5 C	
Zhongjing, Ma et al. (2015) [58]	×	$\times$		$\times$	XB	1000	n/a	15-90%	n/a	
Skugor and Deur (2015) [59]	×		×	×	AB	10	0.92	0-100%	10 kW (0.625 C)	
Wang and Infield (2015) [60]		×		×	XB	42	0.9-0.93	20-100%	-2.16 kW and +2.4 kW	
Nguyen et al. (2015) [61]			×	$\times$	XB	600	n/a	n/a	2.88 kW and 7.68 kW	
Zhang et al. (2017) [62]	$\checkmark$	×	$\checkmark$	×	AB	5000	0.92	$\geq 40\%$	3.3 kW (0.1375 C) and 6.6 kW (0.275 C)	
Peng et al. (2017) [63]		×	×	×	XB	10	n/a	20-95%	10-14 kW (0.4 C-0.625 C)	
Wu et al. (2017) [64]	v		×	×	1EV	1	0.9	20-90%	3.6 kW and 10 kW (0.1 C-0.7 C)	
Mouli et al. (2017) [65]	$\checkmark$	$\checkmark$	$\checkmark$	×	ХВ	6	0.95	5 kW-80%	-10 kW and +50 kW (0.5 C-0.66 C)	
Ramadan et al. (2018) [66]		×	×	×	XB	300	n/a	>40%	4 kW (0.16 C)	
Tavakoli et al. (2018) [67]			×	$\times$	XB	100	(0.9), 0.95	20-100%	4 kW (0.2 C)	
Turker and Colak (2018) [68]			×		1EV	1	n/a	10-100%	10 kW (0.3 C)	
Bucić et al. (2018) [69]		×	×	×	XB	25	(0.75), 0.9	20-90%	3.5 kW (0.35 C), 50 kW (fleet)	
Ahmadian et al. (2018) [70]	$\checkmark$		$\checkmark$	×	XB	15	0.9	dep = 100%	n/a	
Vuelvas et al. (2020) [71]	×	$\times$	×	×	AB	1000	0.9	n/a	3 kW (0.25 C)	
Cheng et al. (2020) [72]	$\checkmark$		×	$\times$	XB	10	n/a	>20%	9.6 kW	
Li et al. (2020) [73]	$\checkmark$	$\checkmark$	$\checkmark$	×	XB	200, 400, 500	n/a	12–94%,	7 kW (0.16 C)	

Table 1. Battery modeling and charging parameters found in the literature.

Table 1.	Cont.
----------	-------

Study	V2G	RES	Degr.	CO <sub>2</sub>	Battery Modeling	Number of EVs	(Dis)Charge Efficiency	SOC Limits	(Max) C-Rate
Das et al. (2020) [74]			$\checkmark$		ХВ	10	0.9	≥20%	3 kW (0.1 C)
Sepetanc et al. (2021) [75]	×	×	v	×	CB	100	0.9-0.93	$\geq 20\%$	1 C
Essiet and Sun (2021) [76]	$\checkmark$	×		$\times$	CB	$5 \times 300$	0.95	10-90%	6.6 kW (0.165 C)

V2G = vehicle-to-grid; RES = renewable energy sources; Degr = electric vehicle battery degradation considered;  $CO_2$  =  $CO_2$  emissions; 1EV = only 1 EV; AB = aggregated battery with a single state-of-charge state variable; CB = cluster-based battery; XB = several individual batteries; SOC = state of charge; n/a = information not available; kW = kilowatt; dep = minimal departure SOC.

The charging and discharging efficiencies of the examined papers, as displayed in Table 1, are approximately 90% with a charging power in the range of about 3–10 kW. The C-rates are in the range of 0.1 to 1 C. In [77], based on several studies examined, it was assumed that a range between 2 C and 5 C was realistic for the maximum C-rate for EVs, yet a higher C-rate increases battery degradation and thus significantly affects the life cycle of a battery (Table 2). For this reason, there are very few studies on optimizing the charging of EVs with very high C-rates. However, the assumed values depend strongly on the vehicle types and the maximum charging rates of the charging stations. SOC limits range from 0 to 100%, often assuming a lower limit of 20% and an upper limit of approximately 85 to 95%. This is first to account for unexpected trips and second to avoid a higher depth-of-discharge (DOD) of the EV battery, as this degrades the battery faster [78]. DOD is the difference in SOC in a cycle (a cycle between 30 and -80% SOC means 50% DOD). The number of EVs simulated depends primarily on the study objective and the battery model. Most studies tend to investigate bidirectional charging in the domestic setting. In the aggregated battery model with only one single-state-of-charge variable, larger fleets of EVs are modeled more often. As V2G and, consequently, load management of EVs are typical demand-side management issues, the charging activity of EVs is influenced by dynamic price-based systems (real-time pricing (RTP), time of use, critical peak pricing, and peak time rebates). According to Amin et al. [79], it appears that most of the research related to EV charging has been conducted in the RTP environment. With this dynamic tariff, the electricity price changes in a regular time interval depending on the grid load to represent the real-time structure. Therefore, the RTP price type seems to be superior to all other systems and provides a solid platform to represent EV charging scenarios in a demand response environment.

### 4.1. Single Aggregate Battery

The introduction of clusters in aggregated models is an approach for numerically efficient solutions in EV fleet planning and optimization. In this technique, contrary to aggregated battery models, multiple flexible and practical strategies can be integrated as EVs are assigned to clusters in self-defined sizes. In [76], the clusters described integer variables to which an arbitrary number of EVs were assigned, facilitating the scaling of large vehicle fleets. Here, the cluster structure consists of 21 clusters ranging from 0% to 100% SOC (5% SOC each). Another model that has low computational complexity, as well as lower real-time communication requirements, is the reduced state-space model proposed in [80]. This model, based on the state-space method, describes aggregated EVs with different connection states and different SOCs and predicts state transitions and control capacity estimation using the Markov state transition method. Cardoso et al. [55] modeled the EV fleet (up to 1080 EVs) using a new EV aggregator model and a stochastic extension for schedule uncertainty with four battery sub-models to analyze investment decisions in a large office building.

To achieve this, the EVs in a fleet are distributed to one of four characteristic states (at home, connected to the microgrid, driving home, and driving to the microgrid) at each time step, where all variables related to the operation of the EV are explicitly computed in each time step. Škugor and Deur [59] modeled a fleet of 10 EVs as a single aggregate battery with only the SOC as a state variable for energy planning. Their dynamic programming approach

provides a globally optimal solution with a relatively low computational load due to the low order of the aggregate battery model. In [62], a heuristic intelligent charging strategy for a V2G fleet with aggregated parameters was created to represent the energy and power constraints of an entire EV fleet (5000 EVs were aggregated). Here, instead of predicting individual load profiles or individual charging requirements, the aggregated energy and power constraints of the V2G fleet were captured and predicted via power load profile predictions, time series and probabilistic predictions to be subsequently implemented using a smart charging strategy. In [71], a virtual battery model was used to represent the energy storage and energy demand flexibility capability of a vehicle fleet. This virtual battery represents the total capacity and total flexibility of this fleet and is represented as a dynamic model with stochastic energy demand characterized by different energy consumption scenarios. When the characteristics of EV owners are grouped into behavioral clusters, EV groups with similar demand profiles can be generated without the need for detailed models or measurements of EVs. This approach allows the aggregate flexibility of a potentially large number of EVs to be described with a simple first-order model without dealing with the complexity of individual participants' behavior.

## 4.2. Individual Modeling

In the second type of battery modeling, where each EV is simulated individually, there are also different optimization approaches. For example, ref. [56] simulated a sequential arrival of 63 EVs whose charging profiles can be rescheduled at any time. Hence, each EV has a different predefined initial state of charge, arrival and departure time, and a desired final state of charge. For a realistic and variable pattern, the arrival times are chosen randomly so that the average arrival frequency is piecewise constant. The dwell time, as well as the desired end state, correspond to a predefined mean value plus a randomly chosen number within a certain adjustable range.

As with the aggregated virtual battery, probabilistic predictions (normal distribution, two-parameter Weibull distribution, Poisson distribution, Monte Carlo simulations) are applied for arrival and departure times, SOCs (e.g., for minimum, maximum, initial, desired), and driving patterns. Another option involves gathered vehicle data from people to create daily activity profiles with hourly load profile data and driving patterns. In most studies, depending on these parameters and conditions, each EV is simulated individually. The capacity and the charging demand of the EVs are regulated in an aggregator, which is responsible for managing and controlling the charging of EVs, in order to obtain overall economic and environmental results with respect to the optimization goal. The aggregator can also contribute to voltage and frequency support, load balancing and reducing power losses of the EV batteries [78].

In addition, the individual battery simulation provides the opportunity for each EV to implement its own optimal charging strategy while considering battery degradation, to maintain stakeholder and/or customer satisfaction. In [58], 1000 EVs were simulated individually, and the trade-off between the total generation cost and the cumulative battery degradation cost as a function of charging power was studied, while renewable energy and the possibility of V2G were not considered. Li et al. [73], on the contrary, simulated an energy management strategy with the deployment of a photovoltaic (PV) system and an additional battery energy storage system (BESS) for a large EV charging station (200, 400, 500 EVs). As in most studies, battery degradation was considered either as a linear model with a fixed degradation rate [60,62,73] or integrated with EV costs [55] in their economic analysis of cost per kWh cycle. As in [70], the degradation cost was solely integrated into the objective functions, while minimizing the battery degradation cost is seldom considered as an optimization process for EV charging. Similarly,  $CO_2$  emissions were reduced by optimizing the increasing penetration rate of renewables by EVs or by optimizing the time-dependent carbon intensity of the national/local energy mix of energy supply caused by the uptake of energy from the grid. However, optimization for  $CO_2$  emissions reduction is rarely explicitly considered as an objective to be optimized in an EV charging approach. Only 3 of the 23 studies in Table 1 considered minimizing  $CO_2$  emissions. In [68], for example, the self-consumption of solar production was maximized with optimal charging of the vehicle. Cardoso et al. [55] used average seasonal  $CO_2$  emission values for the carbon intensity of the grid. In [74], a time series of specific  $CO_2$  emissions was used for each kWh taken from the grid (kg $CO_2/kWh$ ). However, Fleschutz et al. [81] stated that dynamic average electricity mix emission factors can lead to inaccurate results due to the high variance of emission factors of the different fuel types. Therefore, the use of time-varying carbon emission factors instead of annual average carbon emission factors was suggested for a more accurate representation of the emission values in the energy mix. However, if the actual impact of demand response on operational carbon emissions is to be determined, Fleschutz et al. [81] suggested marginal (power plant) emission factors. Through marginal emission factors, the load change is assessed with the carbon intensity of the demand of the marginal power plant for each time step.

## 5. Battery Degradation

There are several types of batteries for EVs, e.g., nickel-metal hydride (Ni-MH) batteries, lead-acid batteries, and sodium–nickel chloride batteries, but most are based on lithium-ion technology. Recent literature reviews [77,82,83] regarding lithium-ion batteries show that lithium-ion batteries are currently and will be the preferred option for modern EVs and PHEVs because they offer several advantages. These include better charging efficiency, higher number of charging cycles, remarkable high-current capability, high specific energy and power, low memory effect, and, most importantly, the continuous decrease in initial cost and advanced manufacturing technology. As an example, Ni-MH batteries have been almost completely replaced by lithium batteries in EVs and PHEVs. The disadvantage of lithium-ion batteries is mainly high operating temperatures, which can have a negative impact on their energetic performance and, in the context of battery degradation, on their lifetime [84].

The main lithium-ion battery types used for EV applications are as follows: lithium ion manganese oxide (LMO), lithium iron phosphate (LFP), lithium nickel manganese cobalt oxide (NMC), and lithium nickel cobalt aluminum oxides (NCA) [85]. In particular, NMC and NCA batteries seem to be the best for powering EVs [86]. Moreover, according to Guo et al. [87], NMC and NCA are preferred as the choice for electric vehicle power supply due to their inherent characteristics, such as higher power and energy density and low cost.

Nevertheless, for the use in EVs, the life cycle and safety characteristics of lithium-ion batteries should be more important than capacity [85]. In [88], the performance parameters of lithium-ion battery cells were compared. The results show that LFP batteries in particular, despite the high cost and low energy and power density, have a long cycle and calendar life with high safety, favoring the integration of EVs in the V2G context.

Since the focus of this paper is not solely on battery degradation in the context of EVs, a complete discussion on the battery structure and chemical functionality is beyond the scope of this paper; interested readers are directed to [78,89]. In this paper, factors leading to a non-linear and complex aging process of lithium-ion batteries are considered. Battery degradation has a direct impact on the lifetime of a battery and, in addition to reversible self-discharge, usually has an irreversible character. One consequence of the aging process is the increase in internal resistance and the decrease in capacity, leading to a decrease in performance over time.

A distinction is made between calendar (chemical) and cyclic (mechanical) aging. Calendrical aging occurs over time during storage, independent of battery usage. Cyclic aging refers to the effects caused by cycling due to the use of the battery. Both processes lead to a decrease in capacity and thus to a reduction in the remaining useful life [90]. When the available capacity or maximum power of the battery is reduced by 20% (sometimes 30%) of its original value, it is generally considered that the battery has reached its end of life (EOL). This corresponds to a state of health (SOH) of 80%. The SOH gives an indication of

the progress of battery degradation, measured as a percentage. It is defined as the quotient of the current maximum capacity at full charge and the original maximum capacity.

## 5.1. Battery Degradation Models

Various types of models exist in the literature for estimating battery degradation. Although other classifications are possible, battery degradation models can be divided into two main categories, namely empirical [91–94] or semi-empirical models [95–100] and physically based statistical/electrochemical models [101–106]. Such models attempt to describe the dependence of battery capacity and resistance on various aging factors. Both electrochemical and empirical approaches have their advantages and disadvantages. Electrochemical models mainly refer to the reaction mechanism inside the battery and thus rely heavily on fundamental theory to understand the detailed internal electrochemical reaction process and reaction intensity of degradation inside the battery [107]. They derive the simulated behavior from known equations that represent the physical behavior [89]. Due to the complexity and increased computational costs of electrochemical models, the question of practicality in smart grid studies arises. The resulting low computational efficiency and the presence of numerous input parameters represent the main drawback of electrochemical models [108]. To address complexity, semi-empirical and empirical models combine theoretical foundations with experimental results to gradually assign appropriate equations and parameters to the models to achieve the best fit to experimental data.

Table 2 shows current battery degradation models from the literature, characterizes them by type of degradation model and battery, by type of degradation (calendar or cycling degradation), and by factors analyzed, and shows the main highlights of the studies examined. The focus in the examination of the degradation models was placed on cycling degradation, as this is crucial in the context of V2G simulation. The EV experiences calendar degradation regardless of V2G use, and since it is the additional degradation caused by V2G use that is of most interest, V2G studies usually leave calendar degradation out of the equation. Moreover, only a few studies simulate longer periods where calendar degradation is an essential factor of overall battery degradation as, for example, in [109]. Here, the authors conducted the simulation assuming a daily charge/discharge, a cell temperature of 25 °C, and a DOD of 10%. The results show that the SOC condition has a significant impact on degradation costs. If the SOC could be maintained at a low level, the overall costs with a DR strategy could be reduced noticeably. Under these assumptions, calendar degradation was found to be several times higher than cyclical degradation. However, periods of several years were considered.

Most of the papers reviewed as well as many studies in the literature regarding battery degradation models use empirical or semi-empirical models due to their ease of use. According to the studies reviewed, empirical and semi-empirical models are the most suitable candidates in the context of battery degradation in smart grid studies for EVs. While they are not as robust as physical models, their ease of integration into existing V2G models and sufficient accuracy make them applicable to many EV battery degradation scenarios. Therefore, empirical models are often used in battery management systems due to their speed, memory space, and numerical convergence [89]. It can also be seen from Table 2 that the parameters of most models are for lithium-iron-phosphate batteries, which are of greatest interest for EVs. The model output of most studies is the capacity loss of the (lithium) battery. Only a few studies investigated other outputs in this context, such as a prediction of battery life or a maximum number of cycles of the EVs. In the literature, a range between 1000 and 3000 cycles is considered realistic [77]. The most frequently investigated factors leading to the model outputs are: C-rate, SOC, battery temperature, DOD, and time (especially for calendar degradation).

Study	Battery Type	Degradation Model	Calendar Life	Cycle Life	Analyzed Factors	Model Output	Highlights
Yang et al. (2019) [101]	NCA (18,650-type cells)	Electrochemical- thermal- mechanical model	No	Yes	C-Rate, Tem- perature	Battery capacity loss	Lower and higher ambient temperatures accelerate capacity degradation. High C-rates can increase active material loss significantly. Capacity loss is, in the initial stage of aging, nearly linear but starts to accelerate after performing hundreds of cycles.
Friesen et al. (2017) [102]	NCM (18,650-type cells)	Electrochemical model	No	Yes	Temperature	Effect of different aging tem- peratures	Slight temperature fluctuations can result in a propagation of electrochemical reactions.
Motapon et al. (2020) [103]	LFP (26,650 type cells), NCM (18,650-type cells)	Physical model	No	Yes	DOD, Tem- perature, C-Rate	Impact on battery capacity	A generic cycle life model to represent the impact of cycle life drivers.
Wang et al. (2016) [95]	NCM (18,650 type cells)	Semi- empirical	Yes	Yes	Temperature, C-Rate, Time	Battery capacity loss	Minor impacts from infrequent V2G services on battery life. Frequency control and peak load reduction do not result in significant degradation.
Petit et al. (2016) [91]	LFP (A123), NCA (Saft VL6P)	Empirical model	Yes	Yes	SOC, Tem- perature, Current	Battery capacity loss	Major differences between both battery types (LFP and NCA).
Omar et al. (2014) [96]	LFP (battery cells)	Semi- empirical	No	Yes	Temperature, C-Rate, DOD	Maximum number of cycles before EOL (80%)	Operating temperature has a great impact on the service life of the battery. High C-rates have a harmful impact on the performance of the battery.
Ouyang et al. (2016) [104]	LFP, LMO+NCM (battery cells)	Mechanistic and prognostic model	No	Yes	C-Rate, Tem- perature	Battery capacity loss	The lifetime of lithium-ion batteries is very temperature dependent; high temperatures can accelerate the degradation significantly. Commercially available batteries with an LFP cathode degraded significantly slower than batteries with a mixed cathode material (LMO + NCM) under the same conditions. Higher C-rates or SOCs result in faster losses of lithium inventory.
Li et al. (2018) [105]	LFP (A123 batteries)	Electrochemical model	Yes	Yes	Current, Tempera- ture	Battery capacity loss	Capacity losses strongly depend on C-rates and SOCs. SOC and temperature are the most significant parameters affecting storage conditions. During the cycle, the total energy throughput, the C-rate, and the temperature have the most influence on the battery capacity loss.
Uddin et al. (2017) [92]	NCA (18,650 type cells)	Empirical model	Yes	Yes	Temperature, SOC, C-Rate	Battery capacity loss	Smart grids can extend EV battery life (compared to cases without V2G).
Dubarry et al. (2017) [93]	NCA (NCR18650B battery cells)	Empirical model	Yes	Yes	Time, SOC, Tempera- ture	Battery capacity loss	Capacity loss is increased by 75% and resistance by 10% with twice-daily V2G. V2G strategies had a negative impact on cell efficiency.The influence of SOC at higher temperatures is more significant.
Chen et al. (2021) [97]	Ni-MH (battery cells)	Semi- empirical model	No	Yes	C-Rate, Tem- perature, DOD	Battery capacity loss	DOD has the greatest impact on battery life.
Johnen et al. (2021) [106]	NCM, LFP (battery cells)	Statistical model	No	Yes	C-Rate, SOC	Predict capacity behavior of battery	Useful model for batteries in second-life settings.
Olmos et al. (2021) [94]	NMC, LFP (battery cells)	Empirical model	No	Yes	Temperature, C-Rate, DOD, SOC	SOH evolu- tion/relations between stress factors and capacity loss	NMC: DOD and temperature accelerate battery degradation the most. LFP: Discharge C-rate is the most limiting degradation stress factor; low degradation dependency on DOD.
Wang et al. (2014) [98]	NCM + LMO (18,650 type cells)	Semi- empirical model	Yes	Yes	Temperature, C-Rate, DOD	Predict battery life for various tempera- tures and rates	High temperatures lead to a higher loss of calendar life, and low temperatures and high C-rates lead to a higher loss of cycle life.
Sarasketa- Zabala et al. (2016) [99]	LFP (26,650-type cells)	Semi- empirical model	Yes	Yes	SOC, Tem- perature, DOD, C-Rate	Lifetime prognosis	The effects of calendar and cyclic degradation can be used in combination for degradation prediction under dynamic and complex operating conditions.
Cordoba- Arenas et al. 2015 [100]	NMC+LMO (battery pouch cells)	Semi- empirical model	No	Yes	SOC, Tem- perature, C-Rate	Predicting battery cycle life	Degradation model to predict battery life under realistic EV scenarios.

# Table 2. Battery degradation models.

NCA = lithium nickel cobalt aluminum oxide; NCM/NMC = lithium nickel cobalt manganese oxide; LFP = lithium iron phosphate battery; LMO = lithium ion manganese oxide; NiMH = nickel-metal hydride battery; EV = electric vehicle; SOC = state of charge; V2G = vehicle-to-grid; DOD = depth of discharge; EOL = end of life; SOH = state of health.

## 5.2. Battery Degradation Parameters

According to the current state of the art (Table 2), the main external stress factors, besides energy throughput that cause and/or exacerbate cyclic degradation of the battery, are the following:

Battery temperature: In general, temperature proves to be the most important stress factor for cyclic battery degradation, where deviations from the typical 25 °C can lead to an accelerated deterioration [89]. The literature reports that both high and low ambient temperatures [101] can accelerate the degradation of lithium-ion batteries. However, it is also shown that the relationship between temperature and SOC can have a moderate impact. For example, ref. [98] showed that the capacity loss at a low C-rate (approx. 0.5 C) is higher with increasing temperature, whereas at a higher C rate (>5 C), low and high temperatures lead to a higher capacity loss. Das also stated that temperature accelerates all chemical reactions leading to degradation [110]. Arrhenius' law is used to model the dependence of battery degradation on battery temperature. For these reasons, the influence of temperature should not be missing in a battery degradation model.

C-rate: A higher charging rate accelerates the degradation, and this effect can be amplified in combination with the other stress factors. A low C-rate leads to a low DOD. In [90], it was shown that at a high C-rate, battery degradation is high at both low and high SOCs. In [98], it was shown that capacity loss increases with higher C-rates and that this effect tends to be larger at high temperatures. Pelletier et al. [78] stated that a C-rate from one and above leads to an exponentially accelerated reduction of the cycle life of a battery.

State of charge (SOC): The literature shows that higher SOC operation accelerates battery degradation or that a low SOC contributes to reducing battery degradation. For example, ref. [105] found that the capacity loss at higher SOC is slightly higher than at low SOC at all temperatures. Accordingly, the average SOC value also influences the degradation. A higher average SOC means a faster degradation rate, and thus the timing and duration of a charge or discharge event can be adjusted to keep the average SOC low.

Depth of discharge (DOD): Some studies suggest that DOD plays a more important role in battery degradation than SOC. In [97], it was the most relevant factor affecting Ni-MH battery life. Therefore, the DOD should be reduced as much as possible, and it is best to operate in a reasonable working range (25~45 °C). Moreover, in [90], it was pointed out that the battery degrades exponentially with increasing DOD, and therefore a DOD greater than 60% should be avoided to significantly extend battery life. In [94], however, there was a huge difference in the DOD dependence on battery technologies. While for lithium nickel manganese cobalt oxides (NMC) batteries, DOD is seen as the most crucial stress factor, for lithium iron phosphate (LFP) batteries, a low impact of DOD can be seen as an essential advantage.

## 6. Optimization Scenarios in Smart Grids Considering Battery Degradation

Several studies have addressed smart grid optimization scenarios that take battery degradation into account. The optimization problems investigated can be divided into two groups: optimization problems with only one objective (single-objective optimization) and multi-objective optimization, in which the simultaneous optimization of individual objectives is allowed. Here, individual objectives can be weighted and then evaluated regarding their importance for the overall problem in a Pareto analysis. In total, 19 research findings related to single- or multi-objective optimization considering battery degradation in smart grids are presented in Table 3.

For this purpose, the objectives, the integration of battery degradation, the components of the power system (EV, BESS, and RES), the time horizon, and the time resolution of the simulation were investigated. From the perspective of the objective function, most of the approaches studied have only one objective, where the cost function is minimized, or profits are maximized. Degradation costs are only integrated into the cost function. In the multi-objective optimization, profits or costs were simulated with either minimizing battery degradation,  $CO_2$  emissions, grid net exchange, or maximizing EV energy for the building.

Another focus was placed on the applied DR program in the studies reviewed. A distinction was made regarding whether the DR program was incentive-based, price-based, or focused on stand-alone demand-side management factors such as increasing self-consumption of renewable energies. It was found that most of the studies included price-based DR programs. Only one study with incentive-based DR was identified in which a bidding system was presented.

Table 3. Optimization scenarios in smart grids considering battery degradation.

Study	Optimization Method	Objectives	Cycling Degradation	DR Program	RES	Energy Storage	Time Horizon	Time Resolution
Eldeeb et al. (2018) [111]	МОО	Maximizing revenues for PV, minimizing battery degradation	Degradation model (constant T, C-rate)	DSM	PV	BESS, BEV	1 day	10 min
Schuller et al. (2014) [112]	SOO	Minimizing cost of EV charging	Constant parameter	РВ	None	BEV	7 days	15 min
Das et al. (2020) [74]	МОО	Minimizing cost, CO <sub>2</sub> emissions, degradation, grid net exchange	Degradation model (T, C-rate, SOC)	PB, DSM	PV	BEV	1 day	15 min
Di Giorgio et al. (2014) [56]	SOO	Minimizing cost of energy consumption and network losses	Proportional depreciation term	PB, DSM	None	BEV	6 a.m.– 11 p.m.	1 h
Van der Meer et al. (2016) [113]	SOO	Minimizing total cost, increasing PV self-consumption	Constant parameter	PB, DSM	PV	BEV	1 day	15 min
Mouli et al. (2017) [65]	SOO	Minimizing total net costs	Constant degradation penalty for V2G service	PB, DSM	PV	BEV	24 h	15 min
Zhongjing et al. (2015) [58]	SOO	Minimizing the system cost	Degradation cost function (charging power)	РВ	None	BEV	1 day	1 h
Iwafune and Ogimoto 2020 [109]	SOO	Minimizing total cost	Degradation model (DOD)	РВ	PV	BEV	1 year	1 h
Maheshwari et al. (2020) [114]	МОО	Maximizing revenue and minimizing degradation	Degradation model (C-rate, SOC)	РВ	None	BESS	1, 7 days	30 min
Miguel A. Ortega- Vazquez (2014) [115]	SOO	Minimizing total costs	Degradation model (DOD)	РВ	None	BEV	1 day	15 min
Pelzer et al. (2016) [116]	SOO	Maximizing profit	Degradation model (SOC, $\Delta$ SOC, voltage)	РВ	None	BESS, EV	1 year	30 min, 1 h
Xu et al. (2018) [117]	SOO	Maximizing profit	Degradation model (constant T, DOD)	РВ	None	BESS	24 h	1 h
Ahmadian et al. (2018) [70]	SOO	Minimizing total cost	Degradation model (C-rate, DOD)	РВ	Wind	BEV	24 h	1 h
Li et al. (2020) [73]	SOO	Minimizing the total net cost of charging vehicles	Linear degradation model (battery cost)	РВ	PV	BESS, EV	24 h	5 min
Haugen et al. (2021) [118]	SOO	Minimizing the total operational costs	Degradation model (constant T, C-rate)	РВ	None	BESS, EV	5 years	1, 60 min
Tchagang and Yoo (2020) [119]	МОО	Maximizing EV energy for building and minimizing battery degradation	Degradation model (SOC limits)	DSM	None	BEV	1 day (repeated for 12 years)	1 h
Terlouw et al. (2019) [120]	моо	Minimizing the operational costs and CO <sub>2</sub> emissions	Linear degradation model (max energy throughput)	PB, IB, DSM	PV	BESS	1 year	1 h
Malya et al. (2021) [54]	SOO	Maximizing profit from price-based energy arbitrage	Degradation model (constant T, SOC, DOD)	PB	None	BEV	1 year	1 h
Soleimani et al. (2021) [121]	SOO	Minimizing the operational costs including battery degradation	Degradation model (DOD, energy throughput)	PB	PV	BEV	24 h	30 min

MOO = multi-objective optimization; SOO = single-objective optimization; RES = renewable energy sources; T = temperature; SOC = state of charge; PV = photovoltaic; BESS = battery energy storage system; BEV = battery electric vehicle; V2G = vehicle-to-grid; DOD = depth of discharge; IB = incentive-based DR; PB = price-based DR; DSM = demand-side management (stand-alone); min = minutes; h = hours.

# 6.1. Single-Objective Optimization

Few smart grid or EV charging studies in the literature deal with degradation costs as a function of several factors such as DOD, SOC, C-rate, or battery temperature. Mostly, the degradation costs are integrated into the cost function as a constant parameter or in a degradation model. For example, refs. [112,113] included battery degradation in their economic analysis as a constant parameter based on the estimated cycle life. In [122], regarding the relationship between minimum cost, feed-in tariff, and degradation, the impact of different fixed degradation rates was investigated to evaluate the profitability of V2G activities. In [65], the EV owner was compensated with a constant amount of money per kWh provided for V2G service. Some studies use a degradation model with only one influencing parameter; others establish a linear relationship via a linear approximation of variables to account for multiple parameters or to simplify the model. For example, after a linear approximation and after a simplification in the studies [109,115,117], only DOD was seen as a variable external influencing factor of cyclic battery degradation. In [73], a linear degradation was assumed, in which the battery degradation costs depended exclusively on the battery costs and the energy throughput. In [70], two influencing parameters in the battery degradation model were included in the cost function via metaheuristic optimization, namely DOD and C-rate. In the degradation model in [118], two influencing parameters (temperature and C-rate) were taken into account after linearization. Malya et al. [54] developed an arbitrage profit model to evaluate the profit that vehicle owners can make by offering their batteries. The profit is calculated based on the electricity price in the German electricity market, considering a battery degradation model from [123]. The model, which calculates cyclic and calendar degradation, takes into account the influencing factors of temperature, DOD, SOC, time, and accumulated degradation to estimate the current degradation. They estimated a profit of 662 €/EV/year for an EV and showed the potential to encourage owners of EVs to participate in V2G services. Accordingly, in the context of EV fleets in industrial or non-residential environments, instead of selling, the electricity can be used at high demand and price peaks to generate even higher revenues.

## 6.2. Multi-Objective Optimization

In multi-objective optimization where one objective is to minimize battery degradation, the significance is marginal if there is only one constant parameter for battery degradation. Therefore, refs. [74,111,114] modeled a battery model with at least one non-linear stress factor to simulate the non-linear behavior of battery degradation. In [111], revenues of the PV-based EV station were maximized while minimizing the degradation of the BESS with non-linear programming. For this purpose, the operating temperature is assumed to be constant in order to calculate the battery degradation as a function of the C-rate and the energy throughput. Maheshwari et al. [114] used a battery degradation model that takes into account the effects of charging rate and SOC on the degradation of a BESS. The influence of temperature on degradation is not taken into account, as the model was developed for BESS, and the storage system was housed in temperature-controlled containers. In [120], the degradation of battery performance was simplified by assuming linear degradation and applying a limit to battery use to ensure that battery life is maintained. Here, the focus was on minimizing operating costs while minimizing  $CO_2$  emissions. In [119], the profit from selling EV energy to building owners was maximized while battery degradation of EVs was minimized. For this purpose, the battery degradation of the EVs only considered SOC as an external stress factor. For the simulation in this study, a different SOC limit was formulated for each of the five EVs.

Only the study of Das et al. [74] was found to use a battery degradation model in a multi-objective technical-economic-environmental optimization approach that represents the effects of three main stress factors (temperature, C-rate, and SOC). In this study, the battery temperature was assumed to be equal to the average daily ambient temperature and treated as an external impact factor that affects the optimization. In addition, the

simultaneous optimization of four objectives was simulated: minimizing energy costs, CO<sub>2</sub> emissions, cyclic battery degradation of EVs, and grid net exchange.

The literature shows that numerous research studies have focused on the development and opportunities of using EVs in smart grids. However, the battery degradation of EVs has not yet been sufficiently considered. Often, as described above, strong simplifications of the degradation model are made for EVs, or only a few influencing parameters are used. Some studies use an additional BESS in the EV charging scenarios. Moreover, the integration of renewables, as can be seen in Table 3, was only considered in 9 of the 19 studies examined (8 PV, 1 wind energy), and along with this, the minimization of CO<sub>2</sub> was rarely considered as an optimization objective.

Another important aspect of comparison is the choice of the time step in the optimization model, as this influences the result of the optimization model. If the EV charging demand is simulated on a minute basis, the optimization will be performed on a minute basis. Haugen et al. [118] highlighted that when simulating on an hourly basis, the peak loads and the potential savings in the electricity tariff are reduced. The study showed that in short time intervals, small amounts of energy can significantly reduce the power peaks. This effect weakened with larger time steps. Furthermore, shorter time steps resulted in a more accurate EV charging demand and thus higher economic precision, while longer time steps resulted in a reduction in computation expense. The optimization models examined usually use time steps of 15, 30, or 60 min, and only a few use shorter time steps. Considering the development of fast-charging stations and the possibility to buy and sell electricity on the spot market on a 15-min basis, the use of shorter time steps shows a higher feasible economic efficiency. Comparing the time horizon, it is noticeable that many simulations consider a rather shorter period of about 24 h. Only 6 of the 19 studies from Table 3 simulated a longer period than 1 month as, for example, in [119], where simplifications were made, and the same cycle of one day was used for 12 years.

## 7. Discussion and Perspectives

The expansion of electric mobility offers new opportunities to minimize environmental impacts while ensuring that the needs of EV owners and stakeholders are met.

For this reason, the literature on EV charging and discharging was reviewed from the perspective of smart grids, V2G, battery degradation, and demand response. The use and profitability of V2G and smart grids are well documented in numerous research articles. This is demonstrated by issues such as energy arbitrage as in [54] to assess the profit vehicle owners could make by offering their batteries and even the positive effect of V2G on the battery as in [53] ("Beneficial V2G"). Therefore, the focus should be on the key components of smart charging and V2G of large fleets to prevent the uncoordinated and simultaneous charging of EV fleets due to grid impacts and economic reasons. The most challenging aspect for researchers is the appropriate integration of EV fleets to ensure grid stability while minimizing battery degradation of EVs and compensating for the economic losses of vehicle owners.

As part of this, many studies simulate the economic and environmental interests of EV owners under battery degradation and often consider renewable energy generation. Empirical, semi-empirical, and physics-based statistical/electrochemical models have been used in modeling battery degradation. Thus, degradation modeling of EV charging scenarios has largely focused on empirical or semi-empirical modeling over physical modeling. Influential battery degradation parameters regarding cycling degradation (battery temperature, C-rate, DOD, SOC) for EV charging scenarios in the literature were highlighted. It was shown that the non-linear behavior of battery degradation, especially in EV charging scenarios, is not fully taken into account. Adapted battery degradation models are needed for different battery types, as the current battery technologies (NMC, NCA, and LFP) have individual strengths and weaknesses. A detailed overview can be found in [85,88].

In the course of multi-objective optimization of EV charging scenarios considering battery degradation and the trade-off of more than two objectives (e.g., energy cost, CO<sub>2</sub>

emissions, compensation of vehicle owners, and battery degradation), there is a clear lack of studies to identify future potentials. Here, the conflicting goals of the different stakeholders should be optimized in a multi-criteria approach. Thus, in an optimal trade-off scenario, where EV owners that make a net profit by providing their EVs may encourage other EV owners to participate in V2G scenarios or even encourage ICEV owners to adopt an EV. Moreover, the social question should be answered: when and under which circumstances would an EV owner allow another stakeholder (e.g., the company where they are employed) to use their battery?

To the best of the authors' knowledge, there are no such multi-objective optimization of EV charging scenarios in the industrial setting that adequately considers battery degradation and demand response. The main research gaps identified in the current literature can be summarized as follows:

- Battery temperature, which is considered an important loading factor, has generally received little attention. Either the temperature was considered constant or simplifications were made assuming that the battery temperature corresponds to the average daily ambient temperature.
- The minimization as well as the trade-off between CO<sub>2</sub> emissions reduction and battery degradation, in contrast to the minimization of system costs, was not sufficiently proven.
- There is a lack of studies that comprehensively consider battery degradation in EV charging scenarios and balance it with financial compensation for EV owners (especially in the non-residential setting). This trade-off between EV owners and companies regarding the benefits to the company and the balance between battery degradation and profit in the form of subsidized or free charging electricity for the EV owner is not appropriately quantified in the literature.
- There are no studies that have investigated non-residential demand response in smart grids, taking into account larger EV fleets and appropriate battery degradation to show economic and environmental perspectives.

The limitations of this study involve the consideration of battery degradation, which should be further explored. In Table 2, all degradation models refer to test cells or batteries that have been partially interconnected to mimic a battery in an EV. Therefore, analysis of a real EV over a longer period of time would be extremely interesting. Current studies on EV battery use and degradation often refer to standardized data (driving patterns, operating times, driving cycles), and thus real-world degradation in EVs could be under-or overestimated.

# 8. Conclusions

The literature on EV battery use and modeling for non-residential EV charging was reviewed in terms of battery degradation, demand response, and optimization in singleand multi-objective simulations. From the literature review, it appears that further studies focusing on multi-objective optimizations in the non-residential setting in the field of EV fleet charging are needed internationally. In addition, future research studies should consider the following three aspects to ensure best practices:

- 1. When considering battery degradation, battery degradation models should be used that take into account several external influencing factors, in particular battery temperature, C-rate, and SOC or DOD. Regarding battery temperature, as an important influencing factor, more accurate assumptions than average daily ambient temperatures should be used, since battery temperature usually increases with time of use.
- 2. Multi-objective optimizations should consider the trade-off between several objectives, namely energy costs, CO<sub>2</sub> emissions, battery degradation, and compensating vehicle owners for using their batteries in non-residential scenarios, given the economic and environmental drivers. In this context, each objective should be compared with an appropriate weighting, e.g., using a game-theoretic analysis to make profitable decisions.

3. Demand response should be expanded in terms of workplace EV charging. To this end, scenarios in non-residential, i.e., industrial and commercial, sectors should be further investigated to find the economic and environmental profitability as well as the willingness of employees to offer their batteries.

Addressing these three aspects will help future-proof studies in the area of EV battery utilization in non-residential demand response and ensure more robust analyses that can be compared to those of future related studies. In addition, a better understanding of battery degradation through statistical analysis and empirical modeling will lead to higher reliability of predictions regarding profitability.

An interesting area of future research would be the different characteristics of lithium batteries in terms of degradation in EVs during driving and cycle or calendar life. Degradation models could be developed based on the analysis of EV batteries to show potential correlations in the degradation of battery types between driving degradation and cycle/calendar degradation. In addition, the motivation for people to participate in V2G contracts in an industrial setting could be investigated with real-life user studies.

Author Contributions: A.L.: Conceptualization, Methodology, Validation, Investigation, Writing original draft preparation, Writing—review and editing, Visualization. M.F.: Conceptualization, Writing—original draft, Writing—review and editing, Investigation. M.D.M.: Supervision, Conceptualization, Writing—review and editing, Project administration. Funding acquisition. All authors have read and agreed to the published version of the manuscript.

**Funding:** This research was performed as part of the MeSSO Research Group at Munster Technological University (MTU). It was funded by the MTU Risam scholarship scheme, Ireland.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

Conflicts of Interest: The authors declare no conflict of interest.

## Nomenclature

AB	aggregated battery
BESS	battery energy storage system
BEV	battery electric vehicle
DOD	depth of discharge
DR	demand response
EOL	end of life
EV	electric vehicle
ICEV	internal combustion engine vehicle
LFP	lithium iron phosphate
LMO	lithium ion manganese oxide
MOO	multi-objective optimization
NCA	lithium nickel cobalt aluminum oxide
NMC	lithium nickel manganese cobalt oxide
NI-MH	nickel–metal hydride
PHEV	plug-in hybrid electric vehicle
PV	photovoltaic
RES	renewable energy sources
RTP	real-time pricing
SOC	state of charge
SOH	state of health
SOO	single-objective optimization
V2G	vehicle-to-grid
ХВ	individual batteries

# References

- 1. European Commission. Fit for 55: Delivering the EU's 2030 Climate Target on the Way to Climate Neutrality. Available online: https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=CELEX:52021DC0550 (accessed on 1 April 2022).
- Climate Action. Progress Made in Cutting Emissions. Available online: https://ec.europa.eu/clima/eu-action/climate-strategiestargets/progress-made-cutting-emissions\_en (accessed on 1 April 2022).
- Yang, L.; Yu, B.; Yang, B.; Chen, H.; Malima, G.; Wei, Y.-M. Life cycle environmental assessment of electric and internal combustion engine vehicles in China. J. Clean. Prod. 2021, 285, 124899. [CrossRef]
- 4. Shafique, M.; Luo, X. Environmental life cycle assessment of battery electric vehicles from the current and future energy mix perspective. *J. Environ. Manag.* 2022, 303, 114050. [CrossRef] [PubMed]
- 5. Shafique, M.; Azam, A.; Rafiq, M.; Luo, X. Life cycle assessment of electric vehicles and internal combustion engine vehicles: A case study of Hong Kong. *Res. Transp. Econ.* **2021**, *91*, 101112. [CrossRef]
- 6. Bekel, K.; Pauliuk, S. Prospective cost and environmental impact assessment of battery and fuel cell electric vehicles in Germany. *Int. J. Life Cycle Assess* **2019**, *24*, 2220–2237. [CrossRef]
- Requia, W.J.; Mohamed, M.; Higgins, C.D.; Arain, A.; Ferguson, M. How clean are electric vehicles? Evidence-based review of the effects of electric mobility on air pollutants, greenhouse gas emissions and human health. *Atmos. Environ.* 2018, 185, 64–77. [CrossRef]
- 8. Bernard, R.M.; Hall, D.; Lutsey, N. *Update on Electric Vehicle Uptake in European Cities*; International Council on Clean Transportation (ICCT): Hungary, Budapest, 2021.
- 9. PricewaterhouseCoopers. Electric Vehicle Sales Review Q3-2021. Available online: https://www.strategyand.pwc.com/de/en/ industries/automotive/electric-vehicle-sales-review-2021-q3.html (accessed on 14 December 2021).
- 10. International Energy Agency (IEA). Global EV Outlook 2021. 2021. Available online: https://www.iea.org/reports/global-evoutlook-2021 (accessed on 14 March 2022).
- 11. Huang, B.; Meijssen, A.G.; Annema, J.A.; Lukszo, Z. Are electric vehicle drivers willing to participate in vehicle-to-grid contracts? A context-dependent stated choice experiment. *Energy Policy* **2021**, *156*, 112410. [CrossRef]
- 12. Busche, I. Ein Beitrag Zur Optimierten Konzeptauslegung Von Fahrzeugen Im Bereich Der Elektromobilität. Ph.D. Thesis, Otto-von-Guericke-Universität Magdeburg, Magdeburg, Germany, 2014.
- 13. Geske, J.; Schumann, D. Willing to participate in vehicle-to-grid (V2G)? Why not! Energy Policy 2018, 120, 392–401. [CrossRef]
- 14. Ravi, S.S.; Aziz, M. Utilization of Electric Vehicles for Vehicle-to-Grid Services: Progress and Perspectives. *Energies* **2022**, *15*, 589. [CrossRef]
- 15. Nissan. Vehicle to Grid | Nissan Electric Cars & Vehicles | Nissan UK. Available online: https://www.nissan.co.uk/range/electriccars-technology/v2g.html (accessed on 4 April 2022).
- 16. Motors, M. Mitsubishi Motors Implement First 'Vehicle to Grid' Pilot on Dutch Market with Mitsubishi Outlander Phev. Available online: https://www.mitsubishi-motors.com/en/newsrelease/2017/detail1082.html (accessed on 4 April 2022).
- 17. The Mobility House. Kia E-Soul | Charging Station, Charging Cable & Installation Service for Kia E-Soul. Available online: https://www.mobilityhouse.com/int\_en/electric-cars/kia/kia-soul-ev.html#kia-soul-ev-charging-information (accessed on 5 April 2022).
- 18. Volkswagen 2022. Bidirektionales Laden: Die Batterie Des E-Autos Als Stromspeicher. Available online: https://www.volkswagen.de/de/elektrofahrzeuge/elektromobilitaet-erleben/elektroauto-technologie/bidirektionales-laden-die-batteriedes-e-autos-als-stromspeicher.html (accessed on 4 April 2022).
- Cardoso, C.A.; Torriti, J.; Lorincz, M. Making demand side response happen: A review of barriers in commercial and public organisations. *Energy Res. Soc. Sci.* 2020, 64, 101443. [CrossRef]
- 20. Eurelectric. Everything You Always Wanted to Know About Demand Response. 2015. Available online: https://cdn.eurelectric. org/media/1940/demand-response-brochure-11-05-final-lr-2015-2501-0002-01-e-h-C783EC17.pdf (accessed on 9 February 2022).
- 21. Zerrahn, A.; Schill, W.-P. Long-run power storage requirements for high shares of renewables: Review and a new model. *Renew. Sustain. Energy Rev.* 2017, *79*, 1518–1534. [CrossRef]
- 22. International Energy Agency (IEA). Net Zero by 2050—A Roadmap for the Global Energy Sector. 2021. Available online: https://www.iea.org/reports/net-zero-by-2050 (accessed on 14 March 2022).
- Phan, Q.a.; Scully, T.; Breen, M.; Murphy, M.D. Facilitating high levels of wind penetration in a smart grid through the optimal utilization of battery storage in microgrids: An analysis of the trade-offs between economic performance and wind generation facilitation. *Energy Convers. Manag.* 2020, 206, 112354. [CrossRef]
- Paterakis, N.G.; Erdinç, O.; Catalão, J.P. An overview of Demand Response: Key-elements and international experience. *Renew. Sustain. Energy Rev.* 2017, 69, 871–891. [CrossRef]
- 25. IndustRE Project Team. Demand Response in Industrial Production (DRIP). Available online: https://webgate.ec.europa.eu/life/ publicWebsite/index.cfm?fuseaction=search.dspPage&n\_proj\_id=4214 (accessed on 9 February 2022).
- 26. IndustRE Project Team. Using the Exibility Potential in Energy Intensive Industries to Facilitate Further Grid Integration of Variable Renewable Energy Sources (IndustRE). Available online: http://www.industre.eu/ (accessed on 9 February 2022).
- 27. DRIvE Project Team. Demand Response Integration tEchnologies (DRIvE) H2020 Project—Unlocking DR Potential. Available online: https://www.h2020-drive.eu/ (accessed on 9 February 2022).

- 28. FlexLast Project Team. The FlexLast Project: Refrigerated Warehouses Store Energy for Smart Energy Grid. Available online: https://www.zurich.ibm.com/flexlast/infographic\_en/ (accessed on 9 February 2022).
- 29. German Federal Ministry of Education and Research. How the Kopernicus Project SynErgie Helps Industry Match Its Electricity Demand to the Supply. Available online: https://www.kopernikus-projekte.de/en/projects/synergie (accessed on 9 February 2022).
- Dena. Pilot Project DSM Bavaria. Available online: https://www.dena.de/en/topics-projects/projects/energy-systems/pilotproject-dsm-bavaria (accessed on 9 February 2022).
- 31. Molavi, H.; Ardehali, M.M. Utility demand response operation considering day-of-use tariff and optimal operation of thermal energy storage system for an industrial building based on particle swarm optimization algorithm. *Energy Build.* **2016**, 127, 920–929. [CrossRef]
- Stoetzer, M.; Hauer, I.; Richter, M.; Styczynski, Z.A. Potential of demand side integration to maximize use of renewable energy sources in Germany. *Appl. Energy* 2015, 146, 344–352. [CrossRef]
- 33. Gils, H.C. Assessment of the theoretical demand response potential in Europe. Energy 2014, 67, 1–18. [CrossRef]
- Golmohamadi, H. Demand-side management in industrial sector: A review of heavy industries. *Renew. Sustain. Energy Rev.* 2022, 156, 111963. [CrossRef]
- 35. Ayón, X.; Gruber, J.K.; Hayes, B.P.; Usaola, J.; Prodanović, M. An optimal day-ahead load scheduling approach based on the flexibility of aggregate demands. *Appl. Energy* **2017**, *198*, 1–11. [CrossRef]
- Heffron, R.; Körner, M.-F.; Wagner, J.; Weibelzahl, M.; Fridgen, G. Industrial demand-side flexibility: A key element of a just energy transition and industrial development. *Appl. Energy* 2020, 269, 115026. [CrossRef]
- 37. Helin, K.; Käki, A.; Zakeri, B.; Lahdelma, R.; Syri, S. Economic potential of industrial demand side management in pulp and paper industry. *Energy* **2017**, *141*, 1681–1694. [CrossRef]
- Finn, P.; Fitzpatrick, C. Demand side management of industrial electricity consumption: Promoting the use of renewable energy through real-time pricing. *Appl. Energy* 2014, 113, 11–21. [CrossRef]
- 39. Ramin, D.; Spinelli, S.; Brusaferri, A. Demand-side management via optimal production scheduling in power-intensive industries: The case of metal casting process. *Appl. Energy* **2018**, *225*, 622–636. [CrossRef]
- Bohlayer, M.; Fleschutz, M.; Braun, M.; Zöttl, G. Energy-intense production-inventory planning with participation in sequential energy markets. *Appl. Energy* 2020, 258, 113954. [CrossRef]
- Aghajani, G.R.; Shayanfar, H.A.; Shayeghi, H. Demand side management in a smart micro-grid in the presence of renewable generation and demand response. *Energy* 2017, 126, 622–637. [CrossRef]
- 42. Kumbartzky, N.; Schacht, M.; Schulz, K.; Werners, B. Optimal operation of a CHP plant participating in the German electricity balancing and day-ahead spot market. *Eur. J. Oper. Res.* 2017, 261, 390–404. [CrossRef]
- 43. Bischi, A.; Taccari, L.; Martelli, E.; Amaldi, E.; Manzolini, G.; Silva, P.; Campanari, S.; Macchi, E. A rolling-horizon optimization algorithm for the long term operational scheduling of cogeneration systems. *Energy* **2019**, *184*, 73–90. [CrossRef]
- 44. Alipour, M.; Mohammadi-ivatloo, B.; Zare, K. Stochastic risk-constrained short-term scheduling of industrial cogeneration systems in the presence of demand response programs. *Appl. Energy* **2014**, *136*, 393–404. [CrossRef]
- 45. Richstein, J.C.; Hosseinioun, S.S. Industrial demand response: How network tariffs and regulation (do not) impact flexibility provision in electricity markets and reserves. *Appl. Energy* **2020**, *278*, 115431. [CrossRef]
- 46. Tsay, C.; Kumar, A.; Flores-Cerrillo, J.; Baldea, M. Optimal demand response scheduling of an industrial air separation unit using data-driven dynamic models. *Comput. Chem. Eng.* **2019**, 126, 22–34. [CrossRef]
- 47. Arteconi, A.; Ciarrocchi, E.; Pan, Q.; Carducci, F.; Comodi, G.; Polonara, F.; Wang, R. Thermal energy storage coupled with PV panels for demand side management of industrial building cooling loads. *Appl. Energy* **2017**, *185*, 1984–1993. [CrossRef]
- 48. Siddiquee, S.M.S.; Howard, B.; Bruton, K.; Brem, A.; O'Sullivan, D.T.J. Progress in Demand Response and It's Industrial Applications. *Front. Energy Res.* **2021**, *9*, 673176. [CrossRef]
- 49. Vahid-Ghavidel, M.; Javadi, M.S.; Gough, M.; Santos, S.F.; Shafie-khah, M.; Catalão, J.P. Demand Response Programs in Multi-Energy Systems: A Review. *Energies* 2020, *13*, 4332. [CrossRef]
- 50. Jordehi, A.R. Optimisation of demand response in electric power systems: A review. *Renew. Sustain. Energy Rev.* 2019, 103, 308–319. [CrossRef]
- 51. Li, X.; Tan, Y.; Liu, X.; Liao, Q.; Sun, B.; Cao, G.; Li, C.; Yang, X.; Wang, Z. A cost-benefit analysis of V2G electric vehicles supporting peak shaving in Shanghai. *Electr. Power Syst. Res.* **2020**, *179*, 106058. [CrossRef]
- 52. Yilmaz, M.; Krein, P.T. Review of benefits and challenges of vehicle-to-grid technology. In Proceedings of the 2012 IEEE Energy Conversion Congress and Exposition (ECCE), Raleigh, NC, USA, 15–20 September 2012; pp. 3082–3089.
- Bentley, E.; Putrus, G.; Lacey, G.; Kotter, R.; Wang, Y.; Das, R.; Ali, Z.; Warmerdam, J. On Beneficial Vehicle-to-Grid (V2G) Services. In Proceedings of the 2021 9th International Conference on Modern Power Systems (MPS), Cluj-Napoca, Romania, 16–17 June 2021; pp. 1–6.
- Malya, P.P.; Fiorini, L.; Rouhani, M.; Aiello, M. Electric vehicles as distribution grid batteries: A reality check. *Energy Inf.* 2021, 4, S2. [CrossRef]
- 55. Cardoso, G.; Stadler, M.; Bozchalui, M.; Sharma, R.; Marnay, C.; Barbosa-Povoa, A.; Ferrao, P. Optimal investment and scheduling of distributed energy resources with uncertainty in electric vehicle driving schedules. *Energy* **2014**, *64*, 17–30. [CrossRef]

- 56. di Giorgio, A.; Liberati, F.; Canale, S. Electric vehicles charging control in a smart grid: A model predictive control approach. *Control Eng. Pract.* **2014**, *22*, 147–162. [CrossRef]
- 57. Liu, N.; Chen, Q.; Liu, J.; Lu, X.; Li, P.; Lei, J.; Zhang, J. A Heuristic Operation Strategy for Commercial Building Microgrids Containing EVs and PV System. *IEEE Trans. Ind. Electron.* **2015**, *62*, 2560–2570. [CrossRef]
- Ma, Z.; Zou, S.; Liu, X. A Distributed Charging Coordination for Large-Scale Plug-In Electric Vehicles Considering Battery Degradation Cost. *IEEE Trans. Contr. Syst. Technol.* 2015, 23, 2044–2052. [CrossRef]
- Škugor, B.; Deur, J. Dynamic programming-based optimisation of charging an electric vehicle fleet system represented by an aggregate battery model. *Energy* 2015, 92, 456–465. [CrossRef]
- 60. Wang, Y.; Infield, D. Optimal Demand Side Response to Real Time Price Signal Using Electric Vehicles. In Proceedings of the International Conference on Renewable Power Generation (RPG 2015), Beijing, China, 17–18 October 2015; p. 6.
- Nguyen, H.N.T.; Zhang, C.; Mahmud, M.A. Optimal Coordination of G2V and V2G to Support Power Grids with High Penetration of Renewable Energy. *IEEE Trans. Transp. Electrif.* 2015, 1, 188–195. [CrossRef]
- Zhang, H.; Hu, Z.; Xu, Z.; Song, Y. Evaluation of Achievable Vehicle-to-Grid Capacity Using Aggregate PEV Model. *IEEE Trans.* Power Syst. 2017, 32, 784–794. [CrossRef]
- 63. Peng, C.; Zou, J.; Lian, L.; Li, L. An optimal dispatching strategy for V2G aggregator participating in supplementary frequency regulation considering EV driving demand and aggregator's benefits. *Appl. Energy* **2017**, *190*, 591–599. [CrossRef]
- 64. Wu, X.; Hu, X.; Teng, Y.; Qian, S.; Cheng, R. Optimal integration of a hybrid solar-battery power source into smart home nanogrid with plug-in electric vehicle. *J. Power Source* **2017**, *363*, 277–283. [CrossRef]
- Mouli, G.R.C.; Kefayati, M.; Baldick, R.; Bauer, P. Integrated PV Charging of EV Fleet Based on Energy Prices, V2G, and Offer of Reserves. *IEEE Trans. Smart Grid* 2019, 10, 1313–1325. [CrossRef]
- Ramadan, H.; Ali, A.; Nour, M.; Farkas, C. Smart Charging and Discharging of Plug-in Electric Vehicles for Peak Shaving and Valley Filling of the Grid Power. In Proceedings of the 2018 Twentieth International Middle East Power Systems Conference (MEPCON), Cairo, Egypt, 18–20 December 2018.
- Tavakoli, M.; Shokridehaki, F.; Marzband, M.; Godina, R.; Pouresmaeil, E. A two stage hierarchical control approach for the optimal energy management in commercial building microgrids based on local wind power and PEVs. *Sustain. Cities Soc.* 2018, 41, 332–340. [CrossRef]
- Turker, H.; Colak, I. Multiobjective optimization of Grid-Photovoltaic-Electric Vehicle Hybrid system in Smart Building with Vehicle-to-Grid (V2G) concept. In Proceedings of the 2018 7th International Conference on Renewable Energy Research and Applications (ICRERA), Paris, France, 14–17 October 2018.
- Bucic, P.; Lesic, V.; Vasak, M. Distributed Optimal Batteries Charging Control for Heterogenous Electric Vehicles Fleet. In Proceedings of the 2018 26th Mediterranean Conference on Control and Automation (MED), Zadar, Croatia, 19–22 June 2018; pp. 837–842.
- Ahmadian, A.; Sedghi, M.; Mohammadi-ivatloo, B.; Elkamel, A.; Golkar, M.A.; Fowler, M. Cost-Benefit Analysis of V2G Implementation in Distribution Networks Considering PEVs Battery Degradation. *IEEE Trans. Sustain. Energy* 2018, 9, 961–970. [CrossRef]
- Vuelvas, J.; Ruiz, F.; Gruosso, G. Energy price forecasting for optimal managing of electric vehicle fleet. *IET Electr. Syst. Transp.* 2020, 10, 401–408. [CrossRef]
- Cheng, P.-H.; Huang, T.-H.; Chien, Y.-W.; Wu, C.-L.; Tai, C.-S.; Fu, L.-C. Demand-side management in residential community realizing sharing economy with bidirectional PEV while additionally considering commercial area. *Int. J. Electr. Power Energy Syst.* 2020, 116, 105512. [CrossRef]
- 73. Li, D.; Zouma, A.; Liao, J.-T.; Yang, H.-T. An energy management strategy with renewable energy and energy storage system for a large electric vehicle charging station. *Etransportation* **2020**, *6*, 100076. [CrossRef]
- Das, R.; Wang, Y.; Putrus, G.; Kotter, R.; Marzband, M.; Herteleer, B.; Warmerdam, J. Multi-objective techno-economicenvironmental optimisation of electric vehicle for energy services. *Appl. Energy* 2020, 257, 113965. [CrossRef]
- Sepetanc, K.; Pandzic, H. A Cluster-Based Model for Charging a Single-Depot Fleet of Electric Vehicles. *IEEE Trans. Smart Grid* 2021, 12, 3339–3352. [CrossRef]
- Essiet, I.O.; Sun, Y. Optimal open-circuit voltage (OCV) model for improved electric vehicle battery state-of-charge in V2G services. *Energy Rep.* 2021, 7, 4348–4359. [CrossRef]
- König, A.; Nicoletti, L.; Schröder, D.; Wolff, S.; Waclaw, A.; Lienkamp, M. An Overview of Parameter and Cost for Battery Electric Vehicles. World Electr. Veh. J. 2021, 12, 21. [CrossRef]
- 78. Pelletier, S.; Jabali, O.; Laporte, G.; Veneroni, M. Battery degradation and behaviour for electric vehicles: Review and numerical analyses of several models. *Transp. Res. Part B Methodol.* **2017**, *103*, 158–187. [CrossRef]
- Amin, A.; Tareen, W.; Usman, M.; Ali, H.; Bari, I.; Horan, B.; Mekhilef, S.; Asif, M.; Ahmed, S.; Mahmood, A. A Review of Optimal Charging Strategy for Electric Vehicles under Dynamic Pricing Schemes in the Distribution Charging Network. *Sustainability* 2020, 12, 10160. [CrossRef]
- Wang, M.; Mu, Y.; Li, F.; Jia, H.; Li, X.; Shi, Q.; Jiang, T. State Space Model of Aggregated Electric Vehicles for Frequency Regulation. IEEE Trans. Smart Grid 2020, 11, 981–994. [CrossRef]
- 81. Fleschutz, M.; Bohlayer, M.; Braun, M.; Henze, G.; Murphy, M.D. The effect of price-based demand response on carbon emissions in European electricity markets: The importance of adequate carbon prices. *Appl. Energy* **2021**, *295*, 117040. [CrossRef]

- 82. Zubi, G.; Dufo-López, R.; Carvalho, M.; Pasaoglu, G. The lithium-ion battery: State of the art and future perspectives. *Renew. Sustain. Energy Rev.* **2018**, *89*, 292–308. [CrossRef]
- 83. Blomgren, G.E. The Development and Future of Lithium Ion Batteries. J. Electrochem. Soc. 2017, 164, A5019–A5025. [CrossRef]
- 84. Iclodean, C.; Varga, B.; Burnete, N.; Cimerdean, D.; Jurchiş, B. Comparison of Different Battery Types for Electric Vehicles. *IOP Conf. Ser. Mater. Sci. Eng.* 2017, 252, 12058. [CrossRef]
- Hannan, M.A.; Hoque, M.M.; Hussain, A.; Yusof, Y.; Ker, P.J. State-of-the-Art and Energy Management System of Lithium-Ion Batteries in Electric Vehicle Applications: Issues and Recommendations. *IEEE Access* 2018, *6*, 19362–19378. [CrossRef]
- Saldaña, G.; Martín, J.I.S.; Zamora, I.; Asensio, F.J.; Oñederra, O. Analysis of the Current Electric Battery Models for Electric Vehicle Simulation. *Energies* 2019, 12, 2750. [CrossRef]
- 87. Guo, J.; Li, Y.; Pedersen, K.; Stroe, D.-I. Lithium-Ion Battery Operation, Degradation, and Aging Mechanism in Electric Vehicles: An Overview. *Energies* **2021**, *14*, 5220. [CrossRef]
- 88. Hesse, H.; Schimpe, M.; Kucevic, D.; Jossen, A. Lithium-Ion Battery Storage for the Grid—A Review of Stationary Battery Storage System Design Tailored for Applications in Modern Power Grids. *Energies* **2017**, *10*, 2107. [CrossRef]
- Edge, J.S.; O'Kane, S.; Prosser, R.; Kirkaldy, N.D.; Patel, A.N.; Hales, A.; Ghosh, A.; Ai, W.; Chen, J.; Yang, J.; et al. Lithium ion battery degradation: What you need to know. *Phys. Chem. Chem. Phys. PCCP* 2021, 23, 8200–8221. [CrossRef]
- Ahmadian, A.; Sedghi, M.; Elkamel, A.; Fowler, M.; Golkar, M.A. Plug-in electric vehicle batteries degradation modeling for smart grid studies: Review, assessment and conceptual framework. *Renew. Sustain. Energy Rev.* 2018, *81*, 2609–2624. [CrossRef]
- 91. Petit, M.; Prada, E.; Sauvant-Moynot, V. Development of an empirical aging model for Li-ion batteries and application to assess the impact of Vehicle-to-Grid strategies on battery lifetime. *Appl. Energy* **2016**, *172*, 398–407. [CrossRef]
- Uddin, K.; Jackson, T.; Widanage, W.D.; Chouchelamane, G.; Jennings, P.A.; Marco, J. On the possibility of extending the lifetime of lithium-ion batteries through optimal V2G facilitated by an integrated vehicle and smart-grid system. *Energy* 2017, 133, 710–722. [CrossRef]
- 93. Dubarry, M.; Devie, A.; McKenzie, K. Durability and reliability of electric vehicle batteries under electric utility grid operations: Bidirectional charging impact analysis. *J. Power Source* **2017**, *358*, 39–49. [CrossRef]
- 94. Olmos, J.; Gandiaga, I.; Saez-de-Ibarra, A.; Larrea, X.; Nieva, T.; Aizpuru, I. Modelling the cycling degradation of Li-ion batteries: Chemistry influenced stress factors. *J. Energy Storage* **2021**, *40*, 102765. [CrossRef]
- 95. Wang, D.; Coignard, J.; Zeng, T.; Zhang, C.; Saxena, S. Quantifying electric vehicle battery degradation from driving vs. vehicle-to-grid services. *J. Power Source* 2016, 332, 193–203. [CrossRef]
- Omar, N.; Monem, M.A.; Firouz, Y.; Salminen, J.; Smekens, J.; Hegazy, O.; Gaulous, H.; Mulder, G.; Van Den Bossche, P.; Coosemans, T.; et al. Lithium iron phosphate based battery—Assessment of the aging parameters and development of cycle life model. *Appl. Energy* 2014, 113, 1575–1585. [CrossRef]
- 97. Chen, X.; Chu, A.; Li, D.; Yuan, Y.; Fan, X.; Deng, Y. Development of the cycling life model of Ni-MH power batteries for hybrid electric vehicles based on real-world operating conditions. *J. Energy Storage* **2021**, *34*, 101999. [CrossRef]
- 98. Wang, J.; Purewal, J.; Liu, P.; Hicks-Garner, J.; Soukazian, S.; Sherman, E.; Sorenson, A.; Vu, L.; Tataria, H.; Verbrugge, M.W. Degradation of lithium ion batteries employing graphite negatives and nickel–cobalt–manganese oxide + spinel manganese oxide positives: Part 1, aging mechanisms and life estimation. *J. Power Source* 2014, 269, 937–948. [CrossRef]
- 99. Sarasketa-Zabala, E.; Martinez-Laserna, E.; Berecibar, M.; Gandiaga, I.; Rodriguez-Martinez, L.M.; Villarreal, I. Realistic lifetime prediction approach for Li-ion batteries. *Appl. Energy* **2016**, *162*, 839–852. [CrossRef]
- Cordoba-Arenas, A.; Onori, S.; Guezennec, Y.; Rizzoni, G. Capacity and power fade cycle-life model for plug-in hybrid electric vehicle lithium-ion battery cells containing blended spinel and layered-oxide positive electrodes. *J. Power Sources* 2015, 278, 473–483. [CrossRef]
- 101. Yang, S.; Hua, Y.; Qiao, D.; Lian, Y.; Pan, Y.; He, Y. A coupled electrochemical-thermal-mechanical degradation modelling approach for lifetime assessment of lithium-ion batteries. *Electrochim. Acta* **2019**, *326*, 134928. [CrossRef]
- Friesen, A.; Mönnighoff, X.; Börner, M.; Haetge, J.; Schappacher, F.M.; Winter, M. Influence of temperature on the aging behavior of 18650-type lithium ion cells: A comprehensive approach combining electrochemical characterization and post-mortem analysis. *J. Power Sources* 2017, 342, 88–97. [CrossRef]
- 103. Motapon, S.N.; Lachance, E.; Dessaint, L.-A.; Al-Haddad, K. A Generic Cycle Life Model for Lithium-Ion Batteries Based on Fatigue Theory and Equivalent Cycle Counting. *IEEE Open J. Ind. Electron. Soc.* **2020**, *1*, 207–217. [CrossRef]
- 104. Ouyang, M.; Feng, X.; Han, X.; Lu, L.; Li, Z.; He, X. A dynamic capacity degradation model and its applications considering varying load for a large format Li-ion battery. *Appl. Energy* **2016**, *165*, 48–59. [CrossRef]
- 105. Li, D.; Danilov, D.; Zwikirsch, B.; Fichtner, M.; Yang, Y.; Eichel, R.-A.; Notten, P.H. Modeling the degradation mechanisms of C6/LiFePO4 batteries. J. Power Source 2018, 375, 106–117. [CrossRef]
- Johnen, M.; Pitzen, S.; Kamps, U.; Kateri, M.; Dechent, P.; Sauer, D.U. Modeling long-term capacity degradation of lithium-ion batteries. J. Energy Storage 2021, 34, 102011. [CrossRef]
- 107. Yao, L.; Xu, S.; Tang, A.; Zhou, F.; Hou, J.; Xiao, Y.; Fu, Z. A Review of Lithium-Ion Battery State of Health Estimation and Prediction Methods. *World Electr. Veh. J.* **2021**, *12*, 113. [CrossRef]
- Tomasov, M.; Kajanova, M.; Bracinik, P.; Motyka, D. Overview of Battery Models for Sustainable Power and Transport Applications. *Transp. Res. Procedia* 2019, 40, 548–555. [CrossRef]

- 109. Iwafune, Y.; Ogimoto, K. Economic Impacts of the Demand Response of Electric Vehicles Considering Battery Degradation. *Energies* **2020**, *13*, 5771. [CrossRef]
- 110. Das, R. Multi-Objective Smart Charge Control of Electric Vehicles. Ph.D. Thesis, Northumbria University, Newcastle upon Tyne, UK, 2020.
- Eldeeb, H.H.; Faddel, S.; Mohammed, O.A. Multi-Objective Optimization Technique for the Operation of Grid tied PV Powered EV Charging Station. *Electr. Power Syst. Res.* 2018, 164, 201–211. [CrossRef]
- 112. Schuller, A.; Dietz, B.; Flath, C.M.; Weinhardt, C. Charging Strategies for Battery Electric Vehicles: Economic Benchmark and V2G Potential. *IEEE Trans. Power Syst.* 2014, *29*, 2014–2022. [CrossRef]
- 113. van der Meer, D.; Mouli, G.R.C.; Mouli, G.M.E.; Elizondo, L.R.; Bauer, P. Energy Management System with PV Power Forecast to Optimally Charge EVs at the Workplace. *IEEE Trans. Ind. Inf.* **2018**, *14*, 311–320. [CrossRef]
- 114. Maheshwari, A.; Paterakis, N.G.; Santarelli, M.; Gibescu, M. Optimizing the operation of energy storage using a non-linear lithium-ion battery degradation model. *Appl. Energy* **2020**, *261*, 114360. [CrossRef]
- 115. Ortega-Vazquez, M.A. Optimal scheduling of electric vehicle charging and vehicle-to-grid services at household level including battery degradation and price uncertainty. *IET Gener. Transm. Distrib.* **2014**, *8*, 1007–1016. [CrossRef]
- Pelzer, D.; Ciechanowicz, D.; Knoll, A. Energy arbitrage through smart scheduling of battery energy storage considering battery degradation and electricity price forecasts. In Proceedings of the 2016 IEEE Innovative Smart Grid Technologies—Asia (ISGT-Asia), Melbourne, Australia, 28 November–1 December 2016; pp. 472–477.
- Xu, B.; Zhao, J.; Zheng, T.; Litvinov, E.; Kirschen, D.S. Factoring the Cycle Aging Cost of Batteries Participating in Electricity Markets. *IEEE Trans. Power Syst.* 2018, 33, 2248–2259. [CrossRef]
- Haugen, E.; Berg, K.; Torsater, B.N.; Korpas, M. Optimisation model with degradation for a battery energy storage system at an EV fast charging station. In Using the Exibility Potential in Energy Intensive Industries to Facilitate Further Grid Integration of Variable Renewable Energy Sources (IndustRE); IEEE PowerTech: Madrid, Spain, 2021; pp. 1–6.
- 119. Tchagang, A.; Yoo, Y. V2B/V2G on Energy Cost and Battery Degradation under Different Driving Scenarios, Peak Shaving, and Frequency Regulations. *World Electr. Veh. J.* **2020**, *11*, 14. [CrossRef]
- Terlouw, T.; AlSkaif, T.; Bauer, C.; van Sark, W. Multi-objective optimization of energy arbitrage in community energy storage systems using different battery technologies. *Appl. Energy* 2019, 239, 356–372. [CrossRef]
- 121. Soleimani, A.; Vahidinasab, V.; Aghaei, J. Enabling Vehicle-to-Grid and Grid-to-Vehicle Transactions via a Robust Home Energy Management System by Considering Battery Aging. In Proceedings of the 2021 International Conference on Smart Energy Systems and Technologies (SEST), Vaasa, Finland, 6–8 September 2021; pp. 1–6.
- 122. Su, X.; Yue, H. Cost Minimization Control for Smart Electric Vehicle Car Parks. In Proceedings of the 2018 24th International Conference on Automation and Computing (ICAC), Newcastle upon Tyne, UK, 6–7 September 2018; pp. 1–6.
- 123. Xu, B.; Oudalov, A.; Ulbig, A.; Andersson, G.; Kirschen, D.S. Modeling of Lithium-Ion Battery Degradation for Cell Life Assessment. *IEEE Trans. Smart Grid* 2018, 9, 1131–1140. [CrossRef]